# UCSF UC San Francisco Previously Published Works

# Title

Alteration in the gut microbiome is associated with changes in bone metabolism after laparoscopic sleeve gastrectomy

# Permalink

https://escholarship.org/uc/item/5qn911bq

**Journal** Journal of Bone and Mineral Research, 39(2)

# ISSN

0884-0431

# Authors

Wu, Karin C McCauley, Kathryn E Lynch, Susan V <u>et al.</u>

# **Publication Date**

2024-03-22

# DOI

10.1093/jbmr/zjad017

Peer reviewed



# Alteration in the gut microbiome is associated with changes in bone metabolism after laparoscopic sleeve gastrectomy

Karin C. Wu<sup>1,2,\*</sup>, Kathryn E. McCauley<sup>1</sup>, Susan V. Lynch<sup>1</sup>, Renuka R. Nayak<sup>1,2</sup>, Nicole J. King<sup>1,2</sup>, Sheena Patel<sup>3</sup>, Tiffany Y. Kim<sup>1,2</sup>, Katherine Condra<sup>1,2</sup>, Doug Fadrosh<sup>1</sup>, Dat Nguyen<sup>4</sup>, Din L. Lin<sup>1</sup>, Kole Lynch<sup>1</sup>, Stanley J. Rogers<sup>5</sup>, Jonathan T. Carter<sup>5</sup>, Andrew M. Posselt<sup>5</sup>, Lygia Stewart<sup>5,6</sup>, Anne L. Schafer<sup>1,2,7</sup>

<sup>1</sup>Department of Medicine, University of California San Francisco, San Francisco, CA 94143, United States

<sup>2</sup>Medical Services, San Francisco Veterans Affairs Health Care System, San Francisco, CA 94121, United States

<sup>3</sup>California Pacific Medical Center Research Institute, San Francisco, CA 94107, United States

<sup>4</sup>The Campbell Family Institute for Breast Cancer Research, Princess Margaret Cancer Centre, University Health Network, Toronto, ON M5G 2M9, Canada

<sup>5</sup>Department of Surgery, University of California San Francisco, San Francisco, CA 94143, United States

<sup>6</sup>Surgical Services, San Francisco Veterans Affairs Health Care System, San Francisco, CA 94121, United States

<sup>7</sup>Department of Epidemiology and Biostatistics, University of California San Francisco, San Francisco, CA 94143, United States

\*Corresponding author: Karin C. Wu, Department of Epidemiology and Biostatistics; University of California, San Francisco, CA 94143, United States (karin.wu @ucsf.edu).

# Abstract

Laparoscopic sleeve gastrectomy (LSG), the most common bariatric surgical procedure, leads to durable weight loss and improves obesityrelated comorbidities. However, it induces abnormalities in bone metabolism. One unexplored potential contributor is the gut microbiome, which influences bone metabolism and is altered after surgery. We characterized the relationship between the gut microbiome and skeletal health in severe obesity and after LSG. In a prospective cohort study, 23 adults with severe obesity underwent skeletal health assessment and stool collection preoperatively and 6 mo after LSG. Gut microbial diversity and composition were characterized using 16S rRNA gene sequencing, and fecal concentrations of short-chain fatty acids (SCFA) were measured with LC-MS/MS. Spearman's correlations and PERMANOVA analyses were applied to assess relationships between the gut microbiome and bone health measures including serum bone turnover markers (Cterminal telopeptide of type 1 collagen [CTx] and procollagen type 1 N-terminal propeptide [P1NP]), areal BMD, intestinal calcium absorption, and calciotropic hormones.

Six months after LSG, CTx and P1NP increased (by median 188% and 61%, P < .01) and femoral neck BMD decreased (mean -3.3%, P < .01). Concurrently, there was a decrease in relative abundance of the phylum *Firmicutes*. Although there were no change in overall microbial diversity or fecal SCFA concentrations after LSG, those with greater within-subject change in gut community microbial composition ( $\beta$ -diversity) postoperatively had greater increases in P1NP level ( $\rho = 0.48$ , P = .02) and greater bone loss at the femoral neck ( $\rho = -0.43$ , P = .04). In addition, within-participant shifts in microbial richness/evenness ( $\alpha$ -diversity) were associated with changes in IGF-1 levels ( $\rho = 0.56$ , P < .01). The lower the postoperative fecal butyrate concentration, the IGF-1 level ( $\rho = 0.43$ , P = .04). Meanwhile, the larger the decrease in butyrate concentration may influence skeletal outcomes postoperatively, and microbial influences on butyrate formation and IGF-1 are possible mechanisms.

Keywords: gut microbiome, obesity, bariatric surgery, bone turnover markers, bone mineral density, DXA

# Lay Summary

Laparoscopic sleeve gastrectomy (LSG), the most common bariatric surgical procedure, is a highly effective treatment for obesity because it produces dramatic weight loss and improves obesity-related medical conditions. However, it also results in abnormalities in bone metabolism. It is important to understand how LSG affects the skeleton, so that bone loss after surgery might be prevented. We studied adult men and women before and 6 mo after LSG, and we explored the relationship between the altered gut bacteria and bone metabolism changes. We found that:

- Those with greater shifts in their gut bacterial composition had more bone loss.
- Butyrate, a metabolite produced by gut bacteria from fermentation of dietary fiber, was associated with less bone breakdown and higher IGF-1 level (a bone-building hormone).

We conclude that changes in the gut bacteria may contribute to the negative skeletal impact of LSG and reduced butyrate production by the gut bacteria leading to lower IGF-1 levels is a possible mechanism.

# **Graphical Abstract**



Sleeve gastrectomy leads to abnormal bone metabolism and induces alteration in the gut microbiome. This altered gut microbiome is associated with changes in bone turnover marker levels and bone loss at the femoral neck 6 mo postoperatively and thus may be a contributor to negative skeletal consequences of sleeve gastrectomy. Microbial influences on butyrate and IGF-1 production are possible mechanisms.

# Introduction

Bariatric surgery has proven to be a highly effective intervention for severe obesity (BMI  $\ge 40 \text{ kg m}^{-2}$ ), resulting in body weight loss of >20%-30% and improvement in obesity-related conditions.<sup>1,2</sup> With the rise of the obesity epidemic, there is a growing demand for bariatric surgery. Laparoscopic sleeve gastrectomy (LSG) has emerged as the most performed procedure due to its perceived safety and nearly comparable weight loss and metabolic benefits compared to the previous "gold-standard" Roux-en-Y gastric bypass (RYGB).<sup>3,4</sup> However, recent studies suggest that like RYGB, LSG has the unintended consequence of inducing abnormalities in bone metabolism, leading to marked increases in bone turnover marker levels and rapid decreases in BMD.<sup>5-9</sup> The skeletal effects are likely multifactorial, with contributing factors including changes in calcium homeostasis and calciotropic hormones, nutritional factors including calcium malabsorption,9 mechanical unloading, and hormonal changes.<sup>10</sup> Alteration in the gut microbiome is an unexplored potential contributor.

There is emerging evidence that the gut microbiome regulates bone metabolism. The human gut microbiome represents a dynamic ecosystem of microbes that are involved in several essential host functions and can regulate bone homeostasis via its effects on nutrient absorption, vitamin synthesis, immunity, and hormone modulation.<sup>11–14</sup> Animal experiments with both germ-free animals and perturbation of microbiota by antibiotics have demonstrated bone mass alteration that is reversed with colonization.<sup>15–18</sup> In both animal and human studies, supplementation with prebiotics or probiotics to alter the gut microbiome has been shown to improve calcium absorption and protect against bone loss.  $^{19-25}$ 

LSG leads to a rapid and sustained effect on the gut microbiome.<sup>26–28</sup> Although inconsistently described between studies, LSG is generally associated with an increase in microbial  $\alpha$ -diversity (richness and evenness) and a decrease in the Firmicutes:Bacteroidetes ratio.<sup>26,27,29</sup> Many members of the phylum Firmicutes ferment dietary undigested carbohydrates and fibers to produce the end-product short-chain fatty acids (SCFA). Indeed, recent studies have also shown a reduction in fecal SCFA concentration after LSG.<sup>30,31</sup> SCFA are known to improve intestinal mucosal integrity and increase calcium absorption, reduce inflammation, influence bone cell differentiation (osteoblasts and osteoclasts), and modulate hormones including IGF-1,<sup>32-34</sup> which has a role in maintenance of bone health in adulthood.<sup>35</sup> Members of the phylum Bacteroidetes are the main contributors to proinflammatory LPS biosynthesis in the human gut microbiome.<sup>36</sup> Prior studies have found an enrichment of *Bacteroides* in postmenopausal women and older men with osteoporosis.<sup>37–39</sup> Therefore, the postoperative changes in the gut microbiome composition and function may have a negative impact on skeletal health, though no study has yet investigated the role of the gut microbiome in bone metabolism after LSG.

The aim of this study was to determine the relationship between the gut microbiome and skeletal health in a cohort of pre- and postmenopausal women and men with severe obesity undergoing LSG. We hypothesized that gut microbiome alterations contribute to LSG's skeletal effects.

# Materials and methods Study design and population

This was an ancillary investigation to a larger prospective pre-post cohort study that examined calcium metabolism and skeletal health after LSG.9 Women and men aged 24 to 70 yr with severe obesity undergoing LSG were recruited from an academic bariatric surgery center (the University of California, San Francisco) between 2016-2021. Participants were eligible if they were scheduled for an upcoming LSG procedure and were excluded if they were perimenopausal (defined as last menses >3 mo but <4 yr ago), had prior bariatric surgery or intestinal malabsorption, had conditions or were taking medications known to impact bone and mineral metabolism (eg, primary hyperparathyroidism, Paget's disease, hyperthyroidism; on osteoporosis pharmacotherapy, glucocorticoids, thiazolidinediones, aromatase inhibitors, or androgen deprivation therapy), had significant comorbid conditions such as liver and kidney failure, used illicit drugs or alcohol >3 drinks/d, or weighed >200 kg (the weight limit for our DXA scanner). To minimize selection bias upon enrollment, extensive chart review was performed to identify all preoperative patients meeting inclusion criteria to assess eligibility.

### Study protocol

Participants in the parent cohort study underwent study measurements within 3 mo before LSG and again 6 and 12 mo postoperatively. LSG was performed using minimally invasive surgical techniques to remove 60% to 80% of the greater curvature of the stomach by staples from approximately 6 cm proximal to the pylorus to about 1 cm below the gastroesophageal junction. Calcium intake and vitamin D status were standardized throughout study participation. Individualized calcium citrate and vitamin D supplements were supplied at least 2 wk prior to the preoperative study measurements to achieve a daily calcium intake (diet + supplement) of approximately 1200 mg  $d^{-1}$  and to target a 250HD level > 30 ng mL<sup>-1</sup>. Calcium intake and 250HD levels were monitored and supplement doses adjusted throughout the study period. Scheduled check-ins with participants were done at 3, 6, 9, and 12 mo postoperatively to minimize loss to follow-up. Real-time data entry with regular audits was performed to ensure full and accurate data capture.

Of the 55 participants who contributed pre- and postoperative data to the larger parent cohort study, this ancillary study included the 23 participants who provided stool samples at both preoperative and 6-mo postoperative time points. For those who agreed to participate in the ancillary study, additional exclusion criteria included antibiotic therapy or regular pre- or probiotic use  $\leq 3$  mo before each specimen collection. A separate consent was obtained for the ancillary study. Participants were instructed to collect the first bowel movement of the day each time. Participants stored their samples in their home freezers ( $-20 \ ^{\circ}$ C) until delivering them on ice to the study team. Samples were then stored at  $-80 \ ^{\circ}$ C until processing.

The protocol was approved by the Institutional Review Board at the University of California, San Francisco, and all participants provided written informed consent. The study was registered at the US National Institutes of Health (ClinicalTrials.gov, NCT02778490).

#### **Study measures** Biochemical assays

Serum samples were collected after an overnight fast and were immediately analyzed for calcium, creatinine, albumin, phosphate, 25OHD, and PTH at a commercial laboratory (Quest Diagnostics). Remaining sera were stored at -80 °C until batch analysis in a central laboratory (Maine Medical Center Research Institute) for measurement of bone turnover marker levels (serum C-terminal cross-linked telopeptide (CTx) and procollagen type 1 N-terminal propeptide (P1NP)), 1,25[OH]<sub>2</sub>D, and IGF-1 with chemiluminescence on an autoanalyzer (iSYS, Immunodiagnostic Systems). The inter- and intra-assay coefficients of variation were 6.0% and 3.2% for CTx, 5.0% and 2.9% for P1NP, 11.1% and 6.4% for 1,25(OH)<sub>2</sub>D, and 5.0% and 2.16% for IGF-1.

## Body composition and skeletal imaging

Weight and height were measured. Waist circumference was measured at the level directly below the lowest rib, and hip circumference at the maximum extension of the buttocks, viewed from the side. BMI (kg m<sup>-2</sup>) was calculated. Areal BMD at the lumbar spine and proximal femur and estimated total and regional body composition were assessed with wholebody DXA (Horizon A, Hologic).<sup>40</sup> If a participant's body dimensions exceeded the scanning field, modified half-body scans were used.<sup>41</sup>

## Gut microbial profiling

DNA was extracted from all stool samples using a modified cetyltrimethylammonium bromide buffer extraction protocol as previously described.<sup>42</sup> Every extraction batch had a minimum of one negative control (PBS) per plate, placed in the middle of the plate. These control extractions were taken through the full sequencing pipeline. PCR amplification of the V4 region of the 16S rRNA gene was amplified using 515F and 806R primer pairs on the Illumina NextSeq 500 Platform on a 153 bp  $\times$  153 bp sequencing run (Illumina, San Diego).<sup>43</sup> Bacterial reads were demultiplexed by barcode using OIIME scripts (v1.9.1).44 Sequence data underwent Divisive Amplicon Denoising Algorithm 2 (DADA2; v1.20) processing within each run.<sup>45</sup> Briefly, sequences were filtered if they had any N's, had a maximum expected error greater than 2, a quality score <2, or matched to the PhiX genome, filtering approximately 20% of sequences per sample. The software then learned errors, dereplicated sequences, performed denoising, merged reads with a minimum overlap of 25 bp, and checked for chimeras. Taxonomic classifications were assigned using the SILVA database v138. Phylogenetic trees were constructed using the phangorn (v2.8.1), msa (v1.24), and ape (v5.5) packages in R (v4.0.3). Sequence variants (SV) associated with negative controls were removed outright if they were present in >15% of negative controls and < 15% of samples; the mean of remaining variants was subtracted from samples. The resulting SV tables were combined and representatively rarefied to 18000 reads per sample, resulting in all samples retained for analysis (Supplementary Figure S1).

 $\alpha$ -Diversity, a within-sample diversity measure, was determined by Shannon diversity index, Faith's phylogenetic diversity, Chao1 index (richness), and Pielou's evenness index using phyloseq package (version 1.34) in R.  $\beta$ -Diversity, a betweensamples community compositional dissimilarity measure, was calculated using Bray-Curtis (highlighting dissimilarities in higher abundance taxa), Canberra distance, and weighted (accounting for abundance of taxa) and unweighted UniFrac distances—both measures of phylogenetic relatedness—with phyloseq (version 1.34) and vegan package (version 2.5-7) in R.

#### Fecal short-chain fatty acids profiling

Dry fecal samples were analyzed for eight SCFA: acetic acid, propionic acid, isobutyric acid, butyric acid, 2-methylbutyric acid, isovaleric acid, valeric acid, and hexanoic acid by LC-MS/MS (Metabolon Method TAM135) at a commercial laboratory (Metabolon, Inc.). The mass spectrometer was operated in negative mode using electrospray ionization. The peak area of the individual analyte product ions was measured against the peak area of the product ions of the corresponding internal standards. Quantitation was performed using a weighted linear least squares regression analysis generated from fortified calibration standards prepared immediately prior to each run. LC-MS/MS raw data were collected and processed using AB SCIEX software Analyst 1.6.3 and processed using SCIEX OS-MQ software v1.7. Data reduction was performed using Microsoft Excel for Office 365 v.16.

#### Statistical methods

Baseline descriptive data were expressed as means  $\pm$  SD or medians (IQR) depending on normality. Spearman's rank correlation was used to characterize the relationships between preoperative microbial diversity ( $\alpha$ -diversity) and study parameters. Adjusted baseline associations were determined with Spearman's partial correlation with age, sex, menopause status, and baseline BMI. Permutational multivariate analysis of variance (PERMANOVA) was employed to evaluate factors that significantly explained variation in microbial  $\beta$ -diversity preoperatively. Adjustment of PERMANOVA with age, sex, menopause status, and baseline BMI were done with adonis2 function in vegan package (version 2.5-7) in R to partition the distance matrix to remove sources of variation.

The paired *t*-test or Wilcoxon signed-rank test was used to examine whether there was a change between preoperative and 6-mo postoperative time points, depending on normality of the variable. Significance level was defined as two-sided *P*-value <.05. Spearman's rank correlation test was used to assess the relationships between the gut microbiome characteristics (changes in  $\alpha$ -diversity,  $\beta$ -diversity, and fecal SCFA pre-post LSG, and abundance of specific taxa/SV at the 6mo postoperative time point) and longitudinal changes in other study parameters. Spearman's partial correlation was performed to evaluate adjusted associations with age, sex, menopause status, and changed in BMI (all in one function). Differentially enriched taxa pre-post LSG were determined with Analysis of Composition of Microbiomes with Bias Correction (ANCOM-BC) method using the ANCOMBC package on GitHub,<sup>46,47</sup> which estimates the unknown sampling fractions and corrects the bias induced by their difference among samples. The predicted functional characteristics of microbial community were performed from the Kyoto Encyclopedia of Genes and Genomes database using Phylogenetic Investigation of Communities by Reconstruction of Unobserved States (PICRUSt2; version 2.5.2), which generates functional predictions of microbial community from amplicon sequences.48 Visualization of PICRUSt2 analysis was performed with the ggpicrust2 package in R (version 1.7.2).49 All

Table 1. Baseline characteristics.

	N=23
Age, yr	$48.0\pm12.0$
Women, <i>n</i> (%)	19 (83)
Premenopausal	10 (53)
Postmenopausal	9 (47)
Race, <i>n</i> (%)	
White	17 (74)
Black/African American	6 (26)
Ethnicity, <i>n</i> (%)	
Hispanic/Latinx	3 (13)
Weight, kg	$116.4 \pm 21.7$
BMI, kg m <sup><math>-2</math></sup>	$42.3 \pm 5.3$
Percent body fat (%)	$43.7 \pm 4.1$
Waist circumference (cm)	$116.0 \pm 11.2$
Waist-hip ratio	$0.88 \pm 0.11$
Diabetes, $n$ (%)	7 (30)
Serum calcium, mg dL <sup>-1</sup>	$9.3 \pm 0.4$
Albumin, g $dL^{-1}$	$4.1 \pm 0.4$
25OHD upon enrollment, ng mL <sup>-1</sup>	$26 \pm 9$
25OHD at pre-op visit, ng mL $^{-1}$	$43 \pm 12$
PTH, pg mL <sup><math>-1</math></sup>	$41 \pm 21$
$1.25(OH)_2D$ , pg mL <sup>-1</sup>	$67 \pm 22$
$CTx. ng mL^{-1}$	0.19 (0.13-0.28)
P1NP. ng m $L^{-1}$	40.9 (34.9-60.7)
Creatinine, mg $dL^{-1}$	$0.77 \pm 0.19$
eGFR, mL min <sup><math>-1</math></sup> 1.73 m <sup><math>-2</math></sup>	$97 \pm 25$
Urine calcium, mg 24 $hr^{-1}$	187 (71-295)
Areal BMD, $g \text{ cm}^{-2}$	
Lumbar spine	$1.118\pm0.130$
Total hip	$1.050\pm0.125$
Femoral neck	$0.875\pm0.129$

Values are means  $\pm$  SD, median (IQR), or count (percentage). 95% reference intervals provided by the test manufacturers: PTH, 11.5-78.4 pg mL^{-1}; CTx, 0.142-1.351 ng mL^{-1} (postmenopausal women); P1NP, 27.7-127.6 ng mL^{-1}; 1,25(OH)\_2D, 15.2-90.1 pg mL^{-1}. Abbreviations: CTx, collagen type 1 C-telopeptide; eGFR, estimated glomerular filtration rate; P1NP, procollagen 1 intact N-terminal propeptide; PTH, intact parathyroid hormone.

statistical analyses were performed using Stata 15.1 software (StataCorp) and R (version 4.0.3).

#### Results

## **Baseline participant characteristics**

Preoperatively, mean age was  $48.0 \pm 12.0$  yr, weight was  $116.4 \pm 21.7$  kg, and BMI was  $42.3 \pm 5.3$  kg m<sup>-2</sup> (Table 1). Of the 23 participants, 19 (83%) were women with 9 (47%) of those postmenopausal and 10 (53%) premenopausal. Mean 25OHD level was  $43 \pm 12$  ng mL<sup>-1</sup> with supplementation at the time of preoperative study measurements. Mean serum calcium, PTH, 1,25(OH)<sub>2</sub>D, serum creatinine, and eGFR levels were within their reference ranges. One of the 9 postmenopausal women and one of the 2 men >50 yr old had an areal BMD T-score between -1.0 and -2.4 at any site preoperatively. None had a T-score  $\leq -2.5$ . None of the premenopausal women or men <50 yr old had an areal BMD Z-score  $\leq -2.0$ . Baseline characteristics of participants in this ancillary study did not differ significantly from the participants in the larger cohort study who did not provide stool samples.

Baseline measures of  $\alpha$ -diversity were positively correlated with bone turnover marker levels (Shannon diversity index: CTx  $\rho$  = +0.45, *P* = .02; P1NP:  $\rho$  = +0.44, *P* = .02 and Chao1

Table 2. Changes in metabolic parameters, calcium homeostasis, and bone metabolism 6 mo after LSG.

N=23	Baseline	6 mo post-op	% Change	Р
Body composition changes				
Weight (kg)	$116.4 \pm 21.7$	$88.4 \pm 18.5$	$-24.0\% \pm 6.8\%$	<.01
BMI (kg $m^{-2}$ )	$42.3 \pm 5.3$	$32.2 \pm 4.9$	$-23.9\% \pm 6.6\%$	<.01
Percentage body fat (%)	$43.7 \pm 4.1$	$36.8 \pm 5.8$	$-16.1\% \pm 9.5\%$	<.01
Waist circumference (cm)	$116.0 \pm 11.2$	$95.9 \pm 9.3$	$-17.1\% \pm 6.7\%$	<.01
Waist-hip ratio	$0.88 \pm 0.11$	$0.85 \pm 0.08$	$-2.2\% \pm 4.4\%$	.02
Laboratory changes				
HbA1c (%)	5.5 (5.1-6.6)	5.0 (4.9-5.4)	-7.8 (-18.2 to -3.7) %	<.01
IGF-1 (ng mL <sup><math>-1</math></sup> )	$127.6 \pm 47.9$	$133.7 \pm 52.3$	$+11.0\% \pm 32.7\%$	.29
Calcium homeostasis				
Serum calcium, mg dL <sup>-1</sup>	$9.3 \pm 0.4$	$9.3 \pm 0.3$	$-0.2\% \pm 4.7\%$	.40
Albumin, g $dL^{-1}$	$4.1 \pm 0.4$	$4.0 \pm 0.3$	$-2.0\% \pm 9.4\%$	.22
25OHD, ng m $L^{-1}$	$43 \pm 12$	$44 \pm 16$	$+8.6\% \pm 42.4\%$	.75
PTH, pg mL <sup><math>-1</math></sup>	$41 \pm 21$	$45 \pm 24$	$+3.8\% \pm 13.0\%$	.18
$1.25(OH)_2D$ , pg mL <sup>-1</sup>	60 (51-75)	84 (72-105)	+37.3 (+13.6 to +72.4) %	<.01
Creatinine, mg $dL^{-1}$	0.75 (0.63-0.92)	0.70 (0.59-0.83)	-9.5 (-17.3 to 0.0) %	.01
eGFR, mL min <sup>-1</sup>	101 (87–116)	108 (95–118)	+3.2(-2.1  to  +13.2)%	.01
Urine calcium, mg 24 hr <sup>-1</sup>	187 (71–295)	178 (92-302)	+7.78 (-53.2 to +163.0) %	.91
Bone turnover markers				
$CTx$ , ng m $L^{-1}$	0.19 (0.13-0.28)	0.50 (0.41-0.83)	+188.4 (+90.8 to +317.6) %	<.01
P1NP, ng m $L^{-1}$	40.9 (34.9-60.7)	72.9 (63.0-92.0)	+61.0 (+32.8 to +109.8) %	<.01
Bone mineral density (DXA)				
Lumbar spine	$1.118 \pm 0.130$	$1.122 \pm 0.120$	$+0.7\% \pm 5.1\%$	.67
Total hip	$1.050 \pm 0.125$	$0.998 \pm 0.132$	$-5.0\% \pm 3.5\%$	<.01
Femoral neck	$0.875\pm0.129$	$0.845\pm0.122$	$-3.3\% \pm 5.4\%$	<.01

Values are means  $\pm$  SDs or median (IQR). *P*-value calculated based on paired *t*-test or Wilcoxon signed-rank based on normality. Abbreviations: CTx, collagen type 1 C-telopeptide; eGFR, estimated glomerular filtration rate; HbA1c, hemoglobin A1c; P1NP, procollagen 1 intact N-terminal propeptide; PTH, intact parathyroid hormone.

index: CTx  $\rho = +0.39$ , P = .04; P1NP  $\rho = +0.54$ , P < .01), that is, those with lower gut microbial richness and evenness had lower CTx and P1NP levels. After adjustment for age, sex, menopause status, and BMI, the associations remained significant (Shannon's diversity index: CTx  $\rho_{\text{partial}} = +0.41$ , P = .05; P1NP  $\rho_{\text{partial}} = +0.45$ , P = .03 and Chao1 index: CTx  $\rho_{\text{partial}} = +0.37$ , P = .07; P1NP  $\rho_{\text{partial}} = +0.63$ , P < .01). The baseline measure of  $\beta$ -diversity (weighted UniFrac) was significantly associated with P1NP (PERMANOVA,  $R^2 = 0.11$ , P < .01). This relationship also remained significant irrespective of adjustment for age, sex, menopause status, and BMI (P < .01).

# Changes in metabolic parameters, calcium homeostasis, and bone metabolism after LSG

Six months after LSG (median 7.5 [6.5-8.2] mo), all participants achieved substantial weight loss, with a mean loss of  $28.0 \pm 9.2$  kg, or a  $24.0\% \pm 6.8\%$  decline from baseline weight (P < .01, Table 2). Glycated hemoglobin (HbA1c) decreased (P < .01) and eGFR improved slightly (P = .01). No statistically significant change in IGF-1 level was observed. All participants maintained a 25OHD level > 25 ng mL<sup>-1</sup> postoperatively without a statistically significant change in serum calcium, PTH, or 24-hr urinary calcium levels. However, 1,25(OH)<sub>2</sub>D level increased from a median of 60 (IQR 51-75) pg mL<sup>-1</sup> to 84 (IQR 72-105) pg mL<sup>-1</sup> (P < .01). Concurrently, bone turnover marker levels increased markedly, by a median of +188.4 (IOR +90.8 to +317.6) % for CTx and +61.0 (IQR +32.8 to +109.8) % for P1NP (P < .01 for both). Areal BMD declined significantly at the total hip (by  $-5.0\% \pm 3.5\%$ , P < .01) and femoral neck  $(-3.3\% \pm 5.4\%, P < .01)$ . There was not a statistically

significant change in lumbar spine areal BMD over the 6-mo period.

#### Changes in the gut microbiome after LSG

Six months after LSG, there was no significant change in  $\alpha$ -diversity (richness and/or evenness within a sample) or  $\beta$ -diversity (overall community microbial composition between samples) compared with paired presurgical samples (Figure 1A and B, Supplementary Figure S2A and B). However, there was a reduction in the relative abundance of bacteria belonging to the phylum *Firmicutes* postoperatively (log2) fold change of -1.08, adjusted P = .02) (Figure 1D). At the genus level, there were microbial taxa/sequence variants that changed significantly pre- and post-LSG (Figure 1E). Notably, there was a significant reduction in the relative abundance of Bifidobacterium bifidum (log2 fold change of -0.87, adjusted P < .001). There was no statistically significant change in dry fecal SCFA concentrations (Figure 1C, Supplementary Figure S2C) or in the functional prediction of microbial SCFA metabolism pathways (Supplementary Figure S3).

Participants with larger shifts in microbial community composition after surgery, as quantified by the Bray-Curtis  $\beta$ -diversity metric, showed greater percentage increase in P1NP levels ( $\rho = +0.48$ , P = .02) and greater bone loss at the femoral neck ( $\rho = -0.43$ , P = .04; Figure 2A and B). The relationships remained significant with the adjustment for age, sex, menopause status, and BMI loss ( $\Delta$ P1NP:  $\rho_{partial} = +0.53$ , P = .02; femoral neck bone loss:  $\rho_{partial} = -0.45$ , P = .05). The association between the change in Bray-Curtis and percentage increase in CTx was not statistically significant ( $\rho = +0.36$ , P = .09; Figure 2C). In addition, participants with greater increases in  $\alpha$ -diversity measures had greater increases in



**Figure 1.** Preoperative and 6-mo postoperative changes in (A)  $\alpha$ -diversity (within-sample diversity) with Shannon's diversity index (other measures shown in Supplementary Figure S2A) using Wilcoxon signed-rank test, (B)  $\beta$ -diversity (between-sample community compositional dissimilarity) with Bray-Curtis (other measures shown in Supplementary Figure S2B) using PERMANOVA, (C) fecal butyrate concentrations (other SCFA shown in Supplementary Figure S2C) using Wilcoxon signed-rank test, (D) relative abundance of microbial taxa at the phylum level by each participant, and (E) all the differentially enriched microbial taxa at the genus level (using ANCOM-BC, adjusted P < .05) arranged in order of log2 fold change, colored by the corresponding phylum level.

IGF-1 levels (Shannon's diversity index:  $\rho = +0.56$ , P < .01; Chao1 index:  $\rho = +0.58$ , P < .01; Figure 3A). The strength of the point estimates for these relationships remained similar with the adjustment for age, sex, menopause status, and BMI loss (Shannon's diversity index:  $\rho_{partial} = +0.45$ , P = .04; Chao1 index:  $\rho_{partial} = +0.39$ , P = .09). Changes in other calciotropic hormones were not associated with postoperative changes in diversity measures.

There were three microbial taxa that were associated with postoperative bone loss. Participants with greater postoperative abundance of *Lachnospiraceae*, which are among the SCFA producers in the human gut, had less femoral neck BMD loss ( $\rho = +0.44$ , P = .03), and participants with greater abundance of *Bacteroides caccae* and *Bacteroides uniformis* postoperatively had greater femoral neck BMD loss ( $\rho = -0.46$  and  $\rho = -0.58$ , P = .03 and .003). The strength of the point estimates for these relationships remained similar with the adjustment for age, sex, menopause status, and BMI loss (*Lachnospiraceae:*  $\rho_{\text{partial}} = +0.42$ , P = .10; *B caccae:*  $\rho_{\text{partial}} = -0.47$ , P = .04; *B uniformis:*  $\rho_{\text{partial}} = -0.63$ , P = .01).

Regarding fecal SCFA concentration, participants with lower butyrate levels postoperatively had lower IGF-1 levels ( $\rho = +0.43$ , P = .04; Figure 3C). Although not statistically significant, participants with greater decreases in butyrate levels tended to have greater increases in CTx levels ( $\rho = -0.39$ , P = .07; Figure 3B). The decrease in butyrate levels



Figure 2. Spearman's c rank correlations between 6-mo postoperative changes in microbial community structure and (A)  $\Delta$  femoral neck BMD, (B)  $\Delta$  P1NP levels, and (C) $\Delta$  CTx levels. The shaded areas represent the 95% confidence intervals for predictions from a linear model.

was negatively associated with the absolute postoperative CTx levels ( $\rho = -0.43$ , P = .04). This relationship strengthened with adjustment for age, sex, menopause status, and BMI loss ( $\rho_{\text{partial}} = -0.50$ , P = .03).

# Discussion

We report the first study to explore the relationship between the gut microbiome and skeletal outcomes in patients undergoing LSG for weight loss. In this cohort study of 23 adults with severe obesity undergoing LSG, we found that greater 6-mo changes in overall microbial composition were associated with greater increases in levels of biochemical markers of bone turnover and greater declines in femoral neck areal BMD. Although there was no overall decline in fecal butyrate levels after LSG, the greater the reduction in butyrate levels, the higher the postoperative level of the bone turnover marker CTx. These findings suggest that LSGinduced alterations in the gut microbiome composition and function could contribute to negative skeletal effects.

Prior to LSG, in the setting of severe obesity, we found an association between gut microbial diversities and biochemical markers of bone turnover. Recent studies report an aberrant gut microbiome in individuals with obesity. The gut microbiome in obesity has a greater ability to metabolize and extract energy from food and to produce SCFA from enhanced fermentation of undigested polysaccharide.<sup>50-52</sup> This functional distinction is reflected by a differential gut microbial composition. The gut microbiome in obesity tends to have a lower microbial diversity, reduced abundances of the Bacteroidetes phylum, and higher abundances of the Firmicutes phylum, which has a dominant role in carbohydrate metabolism.<sup>53–56</sup> These observed obesityassociated gut microbial characteristics and metabolic activities can affect bone metabolism in obesity. We found that those with a lower gut diversity, and thus a gut microbiome that appeared more different than that expected in normal weight, had more suppressed bone turnover marker levels. We postulate that a lower gut microbial diversity in severe obesity may signal a less functionally diverse microbiome with enhanced efficiency in fermentation of substrates and in SCFA production. Preclinical studies have shown that SCFA, in addition to having metabolic effects, decreases bone resorption by inhibiting osteoclast differentiation and thus lowering bone turnover.<sup>34,57</sup> Studies have shown that biochemical markers of bone turnover are lower in adults with obesity compared with normal weight.<sup>58,59</sup> The lower bone turnover may contribute to the seemingly higher BMD in obesity.<sup>58</sup>

The alteration in the gut microbiome we observed 6 mo after LSG is consistent with previous reports.<sup>60-62</sup> We found an overall decrease in Firmicutes with a trend toward an increase in Bacteroidetes phyla without a change in microbial richness 6 mo after LSG. Reportedly, the gut microbiome profile following LSG approaches that of lean individuals and is associated with favorable metabolic outcomes.<sup>63-65</sup> The specific pattern of microbiota changes is unique to LSG compared to RYGB and dietary weight loss intervention.<sup>61,66,67</sup> Possible mechanisms for LSG-induced changes in the gut microbiome include postoperative dietary changes, altered gastrointestinal function and gastric acidity, and hormonal changes. LSG not only leads to decreased food intake but also changes in food choice and frequency of intake. It is known that diet influences the structure and activity of the human gut microbiome rapidly.<sup>68</sup> In addition, the gut microbiome can be shaped by the altered intestinal lumen environment of LSG, including accelerated gastric emptying, increased intestinal motility, and diminished gastric acid production.<sup>69-71</sup> Studies have also suggested that the changes in hormones such as leptin and ghrelin may influence the postoperative gut microbiome, although this is not fully understood.72,73

Six months after LSG, serum bone resorption marker CTx increased by a median of 188.4% and bone formation marker P1NP increased by a median of 61.0%, while DXA-derived areal BMD decreased 5.0% at the total hip and 3.3% at the femoral neck. Many factors likely contribute to BMD loss at the hip after LSG, such as mechanical unloading, hormonal changes, and changes in nutritional factors (vitamin D deficiency, decreased calcium absorption). In the present study, we found that participants with greater postoperative changes in microbial community composition (Bray-Curtis dissimilarly measure) had greater increases in P1NP levels and greater bone loss at the femoral neck. While additional studies demonstrating causality are needed, our correlative findings suggest that gut microbiome alteration may be another contributing factor to LSG's skeletal effects.

The reduction in the gut microbial community's capacity to produce SCFA is a potential mechanism for bone metabolism effects of LSG. After LSG, there was an overall



**Figure 3.** Spearman's rank correlations between 6-mo postoperative (A)  $\Delta \alpha$ -diversity measures (Shannon's diversity index on left, Chao1 index on right) and  $\Delta$  IGF-1 levels, (B)  $\Delta$  fecal butyrate concentrations and  $\Delta$  CTx levels, and (C) fecal butyrate levels and IGF-1 levels. The shaded areas represent the 95% confidence intervals for predictions from a linear model.

reduction in the phylum *Firmicutes* and a notable decrease in Bifidobacterium spp. The main SCFA-producing bacteria in the human gut belong to the phylum Firmicutes, and Bifidobacterium spp are known producers of acetate and lactate, which can be converted into butyrate by other colon bacteria through cross-feeding interactions.<sup>74</sup> Although we did not detect a differential predicted enrichment of microbial genes in SCFA metabolism pathways or find a statistically significant decrease in fecal SCFA concentrations, we found that participants with greater postoperative reduction in fecal butyrate concentration had higher serum levels of CTx after LSG. Concordantly, those with the lowest postoperative abundance of Lachnospiraceae spp, which are among the main SCFA producers in the human gut, had greater femoral neck BMD loss. A decrease in abundance of Lachnospiraceae has been associated with osteoporosis in older adults.<sup>37,75-77</sup>

Given SCFAs' inhibitory effects on osteoclastogenesis and anti-inflammatory capacities,<sup>34,57,78</sup> a reduction in levels may lead to higher bone turnover and more bone loss. It should be noted the limitations with the methods of estimating SCFA production: (1) PICRUSt2 and any amplicon-based functional prediction analysis have limited resolution to distinguish strain-specific metabolic pathways and tend to underestimate metabolic diversity and (2) fecal SCFA concentrations reflect a net outcome of the difference between production and absorption, not a direct indicator of intestinal SCFA production.

SCFA have also been shown to modulate levels of hormones including IGF-1, which promotes bone growth and remodeling.<sup>35,79</sup> Yan and colleagues<sup>80</sup> showed that colonization of adult germ-free mice with microbiota and SCFA supplementation of antibiotic-treated mice both lead

to higher IGF-1 levels, suggesting that microbiota-derived SCFA increase IGF-1 levels to promote skeletal growth and development. Consistent with Yan and colleagues, we found a positive association between fecal butyrate and serum IGF-1 levels after LSG. In addition, participants with greater increases in microbial diversity postoperatively had greater increases in IGF-1. Therefore, we postulate that LSG-induced alterations in gut microbial community structure and its ability to ferment complex nondigestible carbohydrate lead to a decline in SCFA production that can impact bone turnover directly and indirectly via IGF-1.

Strengths of our study include its prospective longitudinal design and its comprehensive assessment of bone and calcium homeostasis. Our cohort was representative of the target patient population undergoing bariatric surgery in the United States. A limitation of our study is the small sample size with heterogeneity in terms of sex, age, menopausal status, and race/ethnicity. The study is limited by its 6-mo duration and the absence of a nonsurgical control group. In addition, 16S rRNA amplicon-based microbiota profiles have limited resolution, are reported as relative rather than absolute quantitation, and do not provide insights into microbial functions. Follow-up studies using strategies to assess microbial function would likely provide further insights into the relationships observed in this study.

In conclusion, this hypothesis-generating, exploratory study suggests that alteration in the gut microbiome may be one factor contributing to the negative skeletal consequences of LSG. A potential mechanism is the reduction in the gut microbial capacity to produce SCFA, which can have direct and indirect effects on bone turnover via IGF-1. Our findings thus offer unique insights into the gut-bone axis. Further investigation into the underlying mechanisms and the development of multifaceted prevention strategies are crucial to mitigate the negative skeletal consequences of LSG.

## Acknowledgments

The authors thank Sarah Palilla, PA, and Elliazar Enriquez, LVN, for their facilitation of study recruitment, and Dolores Shoback, MD, for her expert support.

#### Author contributions

Karin Wu (Conceptualization, Data curation, Formal analysis, Investigation, Methodology, Project administration, Visualization, Writing-original draft, Writing-review & editing), Kathryn E. McCauley (Data curation, Formal analysis, Methodology, Writingreview & editing), Susan V. Lynch (Conceptualization, Investigation, Methodology, Resources, Supervision, Writing-review & editing), Renuka R. Nayak (Investigation, Methodology, Supervision, Writingreview & editing), Nicole King (Data curation, Investigation, Project administration, Writing-review & editing), Sheena Patel (Data curation, Formal analysis, Methodology, Writing-review & editing), Tiffany Kim (Conceptualization, Investigation, Methodology, Project administration, Writing-review & editing), Katherine Condra (Data curation, Investigation, Project administration, Writing-review & editing), Doug Fadrosh (Methodology, Project administration, Resources, Writing-review & editing), Dat Nguyen (Data curation, Methodology, Writing-review & editing), Din Lin (Data curation, Methodology, Writing-review & editing), Kole Lynch (Data curation, Methodology, Writing-review & editing), Stanley Rogers (Conceptualization, Investigation, Writing-review & editing), Jonathan Carter (Conceptualization, Investigation, Writing-review &

editing), Andrew Posselt (Conceptualization, Investigation, Writingreview & editing), Lygia Stewart (Conceptualization, Investigation, Writing-review & editing), Anne Schafer (Conceptualization, Data curation, Funding acquisition, Investigation, Methodology, Project administration, Resources, Supervision, Writing-review & editing)

### Supplementary material

Supplementary material is available at *Journal of Bone and Mineral Research* online.

## Funding

This study was funded by the National Institute of Diabetes and Digestive and Kidney Diseases (NIDDK, R01DK107629). Additional support came from NIDDK (P30DK098722), the National Institute of Arthritis and Musculoskeletal and Skin Diseases (NIAMS, P30AR075055), the National Center for Advancing Translational Sciences (UL1TR001872), the Northern California Institute for Research and Education, and the Benioff Center for Microbiome Medicine at the University of California, San Francisco. K.C.W. has received support from the Department of Veterans Affairs and NIDDK (T32DK00741837). R.R.N. is supported by NIAMS (1K08AR073930). T.Y.K. is supported by a VA Career Development Award (1IK2CX001984). Calcium citrate supplements were donated by Bariatric Advantage. Manuscript content is solely the responsibility of the authors and does not necessarily represent the official views of the National Institutes of Health.

# **Conflicts of interest**

ALS has received investigator-initiated research funding from Amgen. The other authors have nothing to disclose.

# **Data availability**

The data that support the findings of the current study are not publicly available but are available from the corresponding author on reasonable request.

# References

- 1. Buchwald H, Avidor Y, Braunwald E, et al. Bariatric surgery: a systematic review and meta-analysis. *JAMA*. 2004;292(14): 1724–1737.
- Chang SH, Stoll CR, Song J, Varela JE, Eagon CJ, Colditz GA. The effectiveness and risks of bariatric surgery: an updated systematic review and meta-analysis, 2003-2012. *JAMA Surg.* 2014;149(3): 275–287.
- Angrisani L, Santonicola A, Iovino P, et al. Bariatric surgery and endoluminal procedures: IFSO worldwide survey 2014. Obes Surg. 2017;27(9):2279–2289.
- Billeter AT, de la Garza Herrera JR, Scheurlen KM, Nickel F, Billmann F, Muller-Stich BP. MANAGEMENT OF ENDOCRINE DISEASE: which metabolic procedure? Comparing outcomes in sleeve gastrectomy and Roux-en Y gastric bypass. *Eur J Endocrinol*. 2018;179(2):R77–R93.
- Cadart O, Degrandi O, Barnetche T, et al. Long-term effects of Roux-en-Y gastric bypass and sleeve gastrectomy on bone mineral density: a 4-year longitudinal study. *Obes Surg.* 2020;30(9): 3317–3325.
- Bredella MA, Greenblatt LB, Eajazi A, Torriani M, Yu EW. Effects of Roux-en-Y gastric bypass and sleeve gastrectomy on bone mineral density and marrow adipose tissue. *Bone*. 2017;95:85–90.
- 7. Vilarrasa N, de Gordejuela AG, Gomez-Vaquero C, et al. Effect of bariatric surgery on bone mineral density: comparison of

gastric bypass and sleeve gastrectomy. Obes Surg. 2013;23(12): 2086–2091.

- Hofso D, Hillestad TOW, Halvorsen E, et al. Bone mineral density and turnover after sleeve gastrectomy and gastric bypass: a randomized controlled trial (Oseberg). J Clin Endocrinol Metab. 2021;106(2):501–511.
- Wu KC, Cao S, Weaver CM, et al. Intestinal calcium absorption decreases after laparoscopic sleeve gastrectomy despite optimization of vitamin D status. *J Clin Endocrinol Metab.* 2023;108(2): 351–360.
- Gagnon C, Schafer AL. Bone health after bariatric surgery. JBMR Plus. 2018;2(3):121–133.
- Tremaroli V, Backhed F. Functional interactions between the gut microbiota and host metabolism. *Nature*. 2012;489(7415): 242–249.
- Chen YC, Greenbaum J, Shen H, Deng HW. Association between gut microbiota and bone health: potential mechanisms and prospective. J Clin Endocrinol Metab. 2017;102(10): 3635–3646.
- Hernandez CJ, Guss JD, Luna M, Goldring SR. Links between the microbiome and bone. J Bone Miner Res. 2016;31(9):1638–1646.
- Zhang J, Lu Y, Wang Y, Ren X, Han J. The impact of the intestinal microbiome on bone health. *Intractable Rare Dis Res.* 2018;7(3): 148–155.
- 15. Sjogren K, Engdahl C, Henning P, et al. The gut microbiota regulates bone mass in mice. *J Bone Miner Res.* 2012;27(6):1357–1367.
- Cho I, Yamanishi S, Cox L, et al. Antibiotics in early life alter the murine colonic microbiome and adiposity. *Nature*. 2012;488(7413):621–626.
- Schwarzer M, Makki K, Storelli G, et al. *Lactobacillus plantarum* strain maintains growth of infant mice during chronic undernutrition. *Science*. 2016;351(6275):854–857.
- Cox LM, Yamanishi S, Sohn J, et al. Altering the intestinal microbiota during a critical developmental window has lasting metabolic consequences. *Cell*. 2014;158(4):705–721.
- 19. McCabe LR, Irwin R, Schaefer L, Britton RA. Probiotic use decreases intestinal inflammation and increases bone density in healthy male but not female mice. *J Cell Physiol*. 2013;228(8): 1793–1798.
- McCabe L, Britton RA, Parameswaran N. Prebiotic and probiotic regulation of bone health: role of the intestine and its microbiome. *Curr Osteoporos Rep.* 2015;13(6):363–371.
- Britton RA, Irwin R, Quach D, et al. Probiotic *L. reuteri* treatment prevents bone loss in a menopausal ovariectomized mouse model. *J Cell Physiol*. 2014;229(11):1822–1830.
- 22. Lambert MNT, Thybo CB, Lykkeboe S, et al. Combined bioavailable isoflavones and probiotics improve bone status and estrogen metabolism in postmenopausal osteopenic women: a randomized controlled trial. *Am J Clin Nutr.* 2017;106(3): 909–920.
- Li JY, Chassaing B, Tyagi AM, et al. Sex steroid deficiencyassociated bone loss is microbiota dependent and prevented by probiotics. J Clin Invest. 2016;126(6):2049–2063.
- Jafarnejad S, Djafarian K, Fazeli MR, Yekaninejad MS, Rostamian A, Keshavarz SA. Effects of a multispecies probiotic supplement on bone health in osteopenic postmenopausal women: a randomized, double-blind. *Controlled Trial J Am Coll Nutr.* 2017;36(7): 497–506.
- 25. Whisner CM, Martin BR, Nakatsu CH, et al. Soluble corn fiber increases calcium absorption associated with shifts in the gut microbiome: a randomized dose-response trial in free-living pubertal females. J Nutr. 2016;146(7):1298–1306.
- Luijten J, Vugts G, Nieuwenhuijzen GAP, Luyer MDP. The importance of the microbiome in bariatric surgery: a systematic review. *Obes Surg.* 2019;29(7):2338–2349.
- Guo Y, Huang ZP, Liu CQ, Qi L, Sheng Y, Zou DJ. Modulation of the gut microbiome: a systematic review of the effect of bariatric surgery. *Eur J Endocrinol*. 2018;178(1):43–56.

- 28. Palmisano S, Campisciano G, Silvestri M, et al. Changes in gut microbiota composition after bariatric surgery: a new balance to decode. *J Gastrointest Surg.* 2020;24(8):1736–1746.
- Coimbra VOR, Crovesy L, Ribeiro-Alves M, Faller ALK, Mattos F, Rosado EL. Gut microbiota profile in adults undergoing bariatric surgery: a systematic review. *Nutrients*. 2022;14(23):4979. https:// doi.org/10.3390/nu14234979.
- Farup PG, Valeur J. Changes in faecal short-chain fatty acids after weight-loss interventions in subjects with morbid obesity. *Nutrients*. 2020;12(3):802. https://doi.org/10.3390/nu12030802.
- Fujinaga A, Ohta M, Endo Y, et al. Changes of short-chain fatty acids and their receptors in an obese rat model after sleeve gastrectomy. *Obes Surg.* 2022;32(8):2649–2657.
- Tan J, McKenzie C, Potamitis M, Thorburn AN, Mackay CR, Macia L. The role of short-chain fatty acids in health and disease. *Adv Immunol.* 2014;121:91–119.
- Wallimann A, Magrath W, Thompson K, et al. Gut microbialderived short-chain fatty acids and bone: a potential role in fracture healing. *Eur Cell Mater.* 2021;41:454–470.
- Lucas S, Omata Y, Hofmann J, et al. Short-chain fatty acids regulate systemic bone mass and protect from pathological bone loss. *Nat Commun.* 2018;9(1):55.
- 35. Tahimic CG, Wang Y, Bikle DD. Anabolic effects of IGF-1 signaling on the skeleton. *Front Endocrinol (Lausanne)*. 2013;4:6.
- d'Hennezel E, Abubucker S, Murphy LO, Cullen TW. Total lipopolysaccharide from the human gut microbiome silences tolllike receptor signaling. *mSystems*. 2017;2(6):e00046-17. https:// doi.org/10.1128/mSystems.00046-17.
- 37. Wei M, Li C, Dai Y, et al. High-throughput absolute quantification sequencing revealed osteoporosis-related gut microbiota alterations in Han Chinese elderly. *Front Cell Infect Microbiol*. 2021;11:630372.
- Rettedal EA, Ilesanmi-Oyelere BL, Roy NC, Coad J, Kruger MC. The gut microbiome is altered in postmenopausal women with osteoporosis and osteopenia. *JBMR Plus*. 2021;5(3):e10452.
- Akinsuyi OS, Roesch LFW. Meta-analysis reveals compositional and functional microbial changes associated with osteoporosis. *Microbiol Spectr.* 2023;11(3):e0032223.
- Scherzer R, Shen W, Bacchetti P, et al. Comparison of dual-energy X-ray absorptiometry and magnetic resonance imaging-measured adipose tissue depots in HIV-infected and control subjects. *Am J Clin Nutr.* 2008;88(4):1088–1096.
- 41. Tataranni PA, Ravussin E. Use of dual-energy X-ray absorptiometry in obese individuals. *Am J Clin Nutr.* 1995;62(4):730–734.
- DeAngelis KM, Brodie EL, DeSantis TZ, Andersen GL, Lindow SE, Firestone MK. Selective progressive response of soil microbial community to wild oat roots. *ISME J.* 2009;3(2):168–178.
- Caporaso JG, Lauber CL, Walters WA, et al. Ultra-high-throughput microbial community analysis on the Illumina HiSeq and MiSeq platforms. *ISME J.* 2012;6(8):1621–1624.
- Caporaso JG, Kuczynski J, Stombaugh J, et al. QIIME allows analysis of high-throughput community sequencing data. *Nat Methods*. 2010;7(5):335–336.
- Callahan BJ, McMurdie PJ, Rosen MJ, Han AW, Johnson AJ, Holmes SP. DADA2: high-resolution sample inference from Illumina amplicon data. *Nat Methods*. 2016;13(7):581–583.
- 46. Lin H, Peddada SD. Analysis of compositions of microbiomes with bias correction. *Nat Commun.* 2020;11(1):3514.
- Lin H, Eggesbo M, Peddada SD. Linear and nonlinear correlation estimators unveil undescribed taxa interactions in microbiome data. *Nat Commun.* 2022;13(1):4946.
- Douglas GM, Maffei VJ, Zaneveld JR, et al. PICRUSt2 for prediction of metagenome functions. *Nat Biotechnol.* 2020;38(6): 685–688.
- 49. Yang C, Mai J, Cao X, Burberry A, Cominelli F, Zhang L. ggpicrust2: an R package for PICRUSt2 predicted functional profile analysis and visualization. *Bioinformatics*. 2023;39(8):btad470. https://doi.org/10.1093/bioinformatics/btad470.

- 50. Aguirre M, Jonkers DM, Troost FJ, Roeselers G, Venema K. In vitro characterization of the impact of different substrates on metabolite production, energy extraction and composition of gut microbiota from lean and obese subjects. *PLoS One.* 2014;9(11):e113864.
- Schwiertz A, Taras D, Schafer K, et al. Microbiota and SCFA in lean and overweight healthy subjects. *Obesity (Silver Spring)*. 2010;18(1):190–195.
- Turnbaugh PJ, Ley RE, Mahowald MA, Magrini V, Mardis ER, Gordon JI. An obesity-associated gut microbiome with increased capacity for energy harvest. *Nature*. 2006;444(7122):1027–1031.
- Le Chatelier E, Nielsen T, Qin J, et al. Richness of human gut microbiome correlates with metabolic markers. *Nature*. 2013;500(7464):541–546.
- Ley RE, Turnbaugh PJ, Klein S, Gordon JI. Microbial ecology: human gut microbes associated with obesity. Nature. 2006;444(7122):1022–1023.
- 55. Ley RE. Obesity and the human microbiome. *Curr Opin Gastroenterol*. 2010;26(1):5–11.
- Chakraborti CK. New-found link between microbiota and obesity. World J Gastrointest Pathophysiol. 2015;6(4):110–119.
- Montalvany-Antonucci CC, Duffles LF, de Arruda JAA, et al. Short-chain fatty acids and FFAR2 as suppressors of bone resorption. *Bone*. 2019;125:112–121.
- Gkastaris K, Goulis DG, Potoupnis M, Anastasilakis AD, Kapetanos G. Obesity, osteoporosis and bone metabolism. J Musculoskelet Neuronal Interact. 2020;20(3):372–381.
- Evans AL, Paggiosi MA, Eastell R, Walsh JS. Bone density, microstructure and strength in obese and normal weight men and women in younger and older adulthood. J Bone Miner Res. 2015;30(5):920–928.
- Jahansouz C, Staley C, Bernlohr DA, Sadowsky MJ, Khoruts A, Ikramuddin S. Sleeve gastrectomy drives persistent shifts in the gut microbiome. *Surg Obes Relat Dis.* 2017;13(6):916–924.
- Murphy R, Tsai P, Jullig M, Liu A, Plank L, Booth M. Differential changes in gut microbiota after gastric bypass and sleeve gastrectomy bariatric surgery vary according to diabetes remission. *Obes Surg*. 2017;27(4):917–925.
- 62. Shao Y, Shen Q, Hua R, Evers SS, He K, Yao Q. Effects of sleeve gastrectomy on the composition and diurnal oscillation of gut microbiota related to the metabolic improvements. *Surg Obes Relat Dis.* 2018;14(6):731–739.
- Liu R, Hong J, Xu X, et al. Gut microbiome and serum metabolome alterations in obesity and after weight-loss intervention. *Nat Med*. 2017;23(7):859–868.
- 64. Tremaroli V, Karlsson F, Werling M, et al. Roux-en-Y gastric bypass and vertical banded gastroplasty induce long-term changes on the human gut microbiome contributing to fat mass regulation. *Cell Metab.* 2015;22(2):228–238.
- 65. Davies NK, O'Sullivan JM, Plank LD, Murphy R. Altered gut microbiome after bariatric surgery and its association with

metabolic benefits: a systematic review. Surg Obes Relat Dis. 2019;15(4):656-665.

- 66. Tu J, Wang Y, Jin L, Huang W. Bile acids, gut microbiota and metabolic surgery. Front Endocrinol (Lausanne). 2022;13:929530.
- 67. Chen G, Zhuang J, Cui Q, et al. Two bariatric surgical procedures differentially alter the intestinal microbiota in obesity patients. *Obes Surg.* 2020;30(6):2345–2361.
- David LA, Maurice CF, Carmody RN, et al. Diet rapidly and reproducibly alters the human gut microbiome. *Nature*. 2014;505(7484):559–563.
- Braghetto I, Davanzo C, Korn O, et al. Scintigraphic evaluation of gastric emptying in obese patients submitted to sleeve gastrectomy compared to normal subjects. Obes Surg. 2009;19(11):1515–1521.
- Sioka E, Tzovaras G, Perivoliotis K, et al. Impact of laparoscopic sleeve gastrectomy on gastrointestinal motility. *Gastroenterol Res Pract.* 2018;2018:4135813.
- 71. Porat D, Vaynshtein J, Gibori R, et al. Stomach pH before vs. after different bariatric surgery procedures: clinical implications for drug delivery. *Eur J Pharm Biopharm*. 2021;160:152–157.
- Rajala MW, Patterson CM, Opp JS, Foltin SK, Young VB, Myers MG Jr. Leptin acts independently of food intake to modulate gut microbial composition in male mice. *Endocrinology*. 2014;155(3): 748–757.
- 73. Leeuwendaal NK, Cryan JF, Schellekens H. Gut peptides and the microbiome: focus on ghrelin. *Curr Opin Endocrinol Diabetes Obes*. 2021;28(2):243–252.
- 74. Riviere A, Selak M, Lantin D, Leroy F, De Vuyst L. Bifidobacteria and butyrate-producing colon bacteria: importance and strategies for their stimulation in the human gut. *Front Microbiol*. 2016;7:979.
- Das M, Cronin O, Keohane DM, et al. Gut microbiota alterations associated with reduced bone mineral density in older adults. *Rheumatology (Oxford)*. 2019;58(12):2295–2304.
- He J, Xu S, Zhang B, et al. Gut microbiota and metabolite alterations associated with reduced bone mineral density or bone metabolic indexes in postmenopausal osteoporosis. *Aging (Albany* NY). 2020;12(9):8583–8604.
- Ozaki D, Kubota R, Maeno T, Abdelhakim M, Hitosugi N. Association between gut microbiota, bone metabolism, and fracture risk in postmenopausal Japanese women. *Osteoporos Int*. 2021;32(1): 145–156.
- Luu M, Visekruna A. Short-chain fatty acids: bacterial messengers modulating the immunometabolism of T cells. *Eur J Immunol*. 2019;49(6):842–848.
- Yakar S, Rosen CJ, Beamer WG, et al. Circulating levels of IGF-1 directly regulate bone growth and density. J Clin Invest. 2002;110(6):771–781.
- Yan J, Herzog JW, Tsang K, et al. Gut microbiota induce IGF-1 and promote bone formation and growth. *Proc Natl Acad Sci U S A*. 2016;113(47):E7554–E7563.